Author's response to reviews

Title: Evaluation of the in vitro skin permeation of antiviral drugs from penciclovir 1% cream and acyclovir 5% cream used to treat herpes simplex virus infection

Authors:

Nathalie Hasler-Nguyen (nathalie.hasler-nguyen@novartis.com)
Donald Shelton (donald.shelton@novartis.com)
Gilbert Ponard (gilbert.ponard@novartis.com)
Marlene Bader (marlene.bader@novartis.com)
Martina Schaffrik (martina.schaffrik@novartis.com)
Pascal Mallefet (pascal.mallefet@novartis.com)

Version: 6 Date: 31 March 2009

Author's response to reviews: see over
Title: MS: 8370183982091332
Evaluation of the in vitro skin permeation of antiviral drugs from penciclovir 1% cream and acyclovir 5% cream used to treat herpes simplex virus infection

Authors:
Nathalie Hasler-Nguyen: nathalie.hasler-nguyen@novartis.com
Donald Shelton: donald.shelton@novartis.com
Gilbert Ponard: gilbert.ponard@novartis.com
Marlene Bader: marlene.bader@novartis.com
Martina Schaffrik: martina.schaffrik@novartis.com
Pascal Mallefet: pascal.mallefet@novartis.com

Version: 5 Date: 31 March, 2009

Author's response to reviews

To: Scott Edmunds PhD
Senior Editor
BMC-series Journals

Dear Dr Edmunds,

The manuscript was amended according to your formatting changes requested as well as proof reading was done.

We greatly appreciate your cooperation and we are looking forward to reading our manuscript in your journal very soon.

With best regards,

Nathalie Hasler-Nguyen, PhD